Author's response to reviews

Title: Heterotopic pregnancy following ovulation induction by Clomiphene and a healthy live birth: a case report

Authors:

Abbas AH Honarbakhsh (abbas.honarbakhsh@yahoo.com)
Elham EK Khoori (elikami20@yahoo.com)
Simin SM Mousavi (mousavi.simin@yahoo.com)

Version: 6 Date: 26 July 2008

Author's response to reviews:

Dear Chief Editor

Dear Sir

I appreciate the reviewers’ and your comments

I am enclosing with this letter a revised article titled: "Heterotopic pregnancy following ovulation induction by Clomiphene and a healthy live birth: a case report" with addressed the comments.

Everything on the basis of reviewers’ comments was done:

- Referee 1: The article was shortened from 1681 words to 1416 words.

- Referee 2: Some of paragraphs and repetitions were decreased and sonologist was put instead of sonographist and our article was corrected again by native English speakers.

- Referee 3: 1-In the introduction section, paragraph 2 second sentences--“the case ….” was moved to the conclusion section.

2- The last line of the conclusion was changed.

3- The sentence of" epidural …." was deleted and what the referee had recommended was written.

4- The actual CRL was mentioned but we would like to tell the referee about the colour Doppler in fig 3 that we should have done it at that time by pulse Doppler.

5- The reason for doing CS was clarified.
I hope the revised manuscript will conform to the journal style.

We would appreciate if you could study and assess this article again for publishing it in your journal.

Please do not hesitate to contact me if you would have any query.

I look forward to hearing from you.

Yours faithfully,

Elham Khoori